Title: Pancreatic mediastinal pseudocyst presenting as a posterior mediastinal mass with recurrent pleural effusions. A case report and short review of the literature.

Authors:

Vasilis Karamouzos (vkaramouzos@hotmail.com)
Dimitris Karavias (dimitriskaravias@gmail.com)
Dimitris Siagris (dsiagris@yahoo.com)
Christina Kalogeropoulou (chrkalogeropoulou@gmail.com)
Fay Kosmopoulou (faykosmopoulou@gmail.com)
Charalampos Gogos (lambisgogos@gmail.com)
Dimitris Velissaris (dimitrisvelissaris@yahoo.com)

Version: 4
Date: 28 February 2015

Author's response to reviews: see over
Dear Sir or Madam,

We have revised the manuscript and addressed all the reviewers concerns.

In more detail, a literature review was performed, and where was possible newer publications were added. Grammatical and spelling mistakes were corrected and the legends are now further explained. Finally the manuscript is written according to the JMC’s template and the reference style is changed to JMC’s guidelines.

Thank you for taking the time to consider this submission and I look forward to hearing from you.

Yours sincerely

Vasilis Karamouzos